\documentstyle[12pt,twoside]{article}
\input{macros-course}
\begin{document}

\handout{clr}{\mbox{}}{October 17, 1996}

\begin{center}
\large \bf {Readings from} \\ 
\bigskip
\LARGE \bf \it {Introduction to Algorithms,} \\
\bigskip
\large \bf {by T.H. Cormen, C.E. Leiserson and R. L. Rivest}\\
\large \bf {MIT Press and McGraw-Hill, 1989}\\
\large \bf {Chapters 2 and 3.}
\end{center}

\vspace{2in}

\fbox{
\parbox{6in}{
\large{Correction to {\em Stirling's formula}, equation 
$(2.12)$ on page $35$}:
\\
\[
n! = \sqrt{2\pi n}~ ({n\over e})^n~ e^{a_n},~~\mbox{where}~~
{1\over{12n+1}}\leq a_n \leq {1\over 12n}
\]
}}

\end{document}
